Author's response to reviews

Title: Peroxisome proliferator-activated receptor gamma and spermidine/spermine N1-acetyltransferase gene expressions are significantly correlated in human colorectal cancer.

Authors:

Michele Linsalata (m.linsalata@libero.it)
Romina Giannini (romina.giannini@libero.it)
Maria Notarnicola (m.notarnicola@email.it)
Aldo Cavallini (cavaldo@libero.it)

Version: 3 Date: 5 July 2006

Author’s response to reviews:

To BioMed Central Editorial Production Team.

We send to you the revised manuscript with the following changes in the format:

1. the titles from the list of the authors have been removed;
2. the figure (Figure 1) and the tables (Tables 1 and 2) have been referred to in the text.

Yours sincerely

Michele Linsalata, B.D.